Author’s response to reviews

Title: A study protocol of the effectiveness of PEGASUS: a multi-centred study comparing an intervention to promote shared decision making about breast reconstruction with treatment as usual.

Authors:

Diana Harcourt (Diana2.Harcourt@uwe.ac.uk)
Nicole Paraskeva (Nicole.Paraskeva@uwe.ac.uk)
Paul White (Paul.White@uwe.ac.uk)
Jane Powell (Jane.Powell@uwe.ac.uk)
Alexandra Clarke (alexmacleodclarke@gmail.com)

Version: 1 Date: 13 Sep 2017

Author’s response to reviews:

1. Please provide all author email addresses in the title page.

All author email addresses are now in the title page.

2. After the TRN please provide the date of the registration of the trial.

The date of the registration of the trial has been added.

3. In the 'Ethics approval and consent to participate' section of the declarations please state how consent will be obtained from the participants. You should confirm whether this will be informed and whether it will be written or verbal.

Information regarding how informed consent will be obtained has been added.

4. Please provide a list of abbreviations after the discussion section.

A list of abbreviations has been added after the discussion section.
5. We require that protocols of clinical trials follow the SPIRIT guidelines. Please include the SPIRIT flow diagram as a figure within the manuscript and please provide the populated checklist as an additional file. Further details, including a link to the checklist, can be found here: http://www.spirit-statement.org/spirit-statement/

The SPIRIT flow diagram and checklist have been added.

6. Please include a list of figure legends at the end of the main text (after the references) and ensure that no figure titles or legends are included in the figure files.

This has been adhered to.